As shown in \cite{knuth1984} and \cite{lamport1994, mittelbach2004}.

A citation with notes: \citep[see][p.~5]{knuth1984}.

See \autoref{sec:intro} and equation~\eqref{eq:main}.

\section{Intro}\label{sec:intro}
The equation \ref{eq:main} is referenced above.
